```
Mauna Loa Solar Observatory Observer's Log
    ______
       Wed Apr 22 17:01:09 GMT 2009
Year: 09 Doy: 112
Observer: koon
WEATHER COMMENT: Wed Apr 22 17:01:11 GMT 2009
Hazy sky, scattered cirrus and lenticular clouds, wind=20mph from the SW,
temp=42F.
end
Wed Apr 22 17:05:07 GMT 2009
                              CHIP
                                       Start Patrol
Wed Apr 22 17:05:38 GMT 2009
                              PICS
                                       Start Patrol
PSPT COMMENT: Wed Apr 22 17:22:30 GMT 2009
obs
 end
WEATHER COMMENT: Wed Apr 22 17:53:04 GMT 2009
Thin cirrostratus has moved in.
end
Wed Apr 22 18:00:29 GMT 2009
                              PICS
                                       Flat
Wed Apr 22 18:02:39 GMT 2009
                              CHIP
                                       LSD
Wed Apr 22 18:03:01 GMT 2009
                              PICS
                                       End Flat
Wed Apr 22 18:03:07 GMT 2009
                              PICS
                                       ReStart Patrol
Wed Apr 22 18:04:03 GMT 2009
                              CHIP
                                       End LSD
Wed Apr 22 18:04:11 GMT 2009
                              CHIP
                                       BiasLSD
Wed Apr 22 18:04:55 GMT 2009
                                       End BiasLSD
                              CHIP
Wed Apr 22 18:05:02 GMT 2009
                              CHIP
                                       Bias
Wed Apr 22 18:05:41 GMT 2009
                                       End Bias
                              CHIP
Wed Apr 22 18:05:46 GMT 2009
                              CHIP
                                       ReStart Patrol
Wed Apr 22 18:20:07 GMT 2009
                              MKIV
                                      Start Correct
WEATHER COMMENT: Wed Apr 22 18:20:38 GMT 2009
Cirrus gone, sky still hazy.
end
Wed Apr 22 18:21:51 GMT 2009
                              MKIV
                                      End Correct
Wed Apr 22 18:58:23 GMT 2009
                              MKIV
                                      Start Correct
Wed Apr 22 19:00:06 GMT 2009
                              MKIV
                                      End Correct
Wed Apr 22 19:08:36 GMT 2009
                              MKIV
                                      Start Correct
Wed Apr 22 19:10:20 GMT 2009
                              MKIV
                                      End Correct
WEATHER COMMENT: Wed Apr 22 19:43:17 GMT 2009
Cirrostratus has moved in.
  end
**GONG PROBLEM**: Wed Apr 22 21:41:40 GMT 2009
Just got back from about an hour down at the shelter working on some
things, replaced the VME-SG board and reseated a few cables, got it all
back up and running normally again.
```

COMMENT: Wed Apr 22 21:43:22 GMT 2009

Dome blocked instruments while I was down at GONG shelter, repointed and reset all.

____end___

WEATHER COMMENT: Wed Apr 22 21:44:19 GMT 2009

Cirrostratus has thickened.

____end___

NICE PICSLIMB IMAGE: 1919 NICE PICSDISC IMAGE: 1726 NICE CHIP IMAGE: 1723

Wed Apr 22 22:31:34 GMT 2009 CHIP End Patrol Wed Apr 22 22:32:04 GMT 2009 PICS End Patrol

Wed Apr 22 22:50:57 GMT 2009

MkIV